Author’s response to reviews

Title: Effect of methylene blue on the genomic response to reperfusion injury induced by cardiac arrest and cardiopulmonary resuscitation in porcine brain

Authors:

Cecile Martijn (cecile.martijn@surgsci.uu.se)
Lars Wiklund (lars.wiklund@akademiska.se)

Version: 3 Date: 29 June 2010

Author’s response to reviews:

In this latest version of the manuscript (revised(2)) we have:

- included the major required revision which consisted in switching back table 1 and 2 in the right order.

- added the missing details for references 4 and 50 in the required style using EndNote

- ensured that mention of the figures was consistent (Fig 3a was changed to Figure 3A and Figure 3b to Figure 3B).

- spell- and grammar-checked the text

However we did not further crop our figures and would be grateful if you could make this change for us.